Author’s response to reviews

Title: Application value of nomogram and prognostic factors of gastric cancer patients who underwent D2 radical lymphadenectomy

Authors:

Guang-Chuan Mu (muge0123654@sina.com)
Yuan Huang (huangy02354168@163.com)
Zhi-Ming Liu (liuzm036984125@sina.com)
Xiang-Hua Wu (wuxh2036984@sina.com)
Xin-Gan Qin (qinxg5320147@sina.com)
Zhi-Bai Chen (chenzb6520314@sina.com)

Version: 2 Date: 10 Sep 2019

Author’s response to reviews:

Reply to reviewers’ comments

Comments 1: Amy Tyberg, MD (Reviewer 3): Excellent revisions. I appreciate the authors responses to my comments. My only additional comment is that some of the limitations described in the responses could be mentioned in the discussion, specifically the small sample size of non-adenocarcinoma tumors and the fact that negative margins was not assessed as a data point.
Reply: Thanks for your suggestion! we have added the limitation of this study in the discussion section.